Author’s response to reviews

Title: Clinical studies on core-carrier obturation: a systematic review and meta-analysis

Authors:

Wai-Yee Amy Wong (drawong@hku.hk)
Shinan Zhang (zhangsn2000@gmail.com)
Kar-Yan Samantha Li (skyli@hku.hk)
Chengfei Zhang (zhangcf@hku.hk)
Chun-Hung Chu (chchu@hku.hk)

Version: 2 Date: 29 Apr 2017

Author’s response to reviews:

Prof Mariana Braga,
Associate Editor,
BMC Oral Health BioMed Central Ltd.

29 April 2017

Dear Prof Mariana Braga,

Re: Submission of revised manuscript OHEA-D-16-00222R1

Thank you very much for your review of our manuscript entitled “Clinical studies on core-carrier obturation: a systematic review and meta-analysis”.

We have revised the manuscripts according to the editor’s suggestions and comments, and a point-to-point response is attached for your reference.
We affirm that we have no financial affiliation (e.g., employment, direct payment, stock holdings, retainers, consultancies, patent licensing arrangements or honoraria), or involvement with any commercial organization with direct financial interests in the subject or materials discussed in this manuscript, nor have any such arrangements existed in the past three years. Any other potential conflict of interest is disclosed.

Thank you very much for your attention.

Yours faithfully,

Professor Chun-Hung Chu

Corresponding author